Author's response to reviews

Title: Overexpression of members of the microRNA-183 family is a risk factor for lung cancer: a case control study

Authors:

Zhu Wangyu (zhuwangyu24@gmail.com)
Liu Xiaoguang (lxg2004@gmail.com)
He Jianying (hjyncmcstudent@163.com)
Chen Dongdong (wydxzhishi@sina.cn)
Huang Yanyan (huangyanyan2005@sina.cn)
Zhang Yongkui (zyk801801@126.com)

Version: 4 Date: 6 September 2011

Author's response to reviews: see over
Dear Editor Danrolf,

Thank you very much for your advice. We have addressed the comments raised by the editor.

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)

We wrote the name in the style of 'First name Initial Last name'.

Authors' contributions: Please use initials instead of full name.
Authors' contributions: Please indicate that all authors read and approved the final manuscript.

References: Please make sure references are cited in numerical order. Currently they are not. Reference 6 is cited after reference 32.

We checked the references again.

Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.

We highlighted the shaded text by making it bold in the tables.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes
 above)

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

**We checked the typographical errors and formats of the manuscript.**

We hope the paper will be published early.

Thank you very much.

With best wishes,

Yours sincerely,

Wangyu Zhu

Corresponding author: Yongkui Zhang

Department of Cardio-Thoracic Surgery; Zhoushan Hospital of Zhejiang Province, Zhoushan, Zhejiang, 316004, China;

Fax: +86-580-2558005; Phone: +86-580-2558362

E-mail:  [zyk801801@126.com](mailto:zyk801801@126.com), zhuwangyu24@gmail.com